Author’s response to reviews

Title: BMI and Psychological Distress in 68,000 Swedish Adults. A weak association when controlling for an age-gender combination.

Authors:

Susanne Brandheim (susanne.brandheim@kau.se)
Ulla Rantakeisu (ulla.rantakeisu@kau.se)
Bengt Starrin (bengt.starrin@kau.se)

Version: 4 Date: 11 January 2013

Author’s response to reviews:

There was only one revision needed, to include a proper title page to the manuscript. This has been done. Thankyou for all input.